Debugging locals Print the lo cal va riables of the selected frames stack Print backtrace of the entire stack Debugging Command buttons run Begin rogram execution cont Continue execution from where it stopp ed next Execute one source line, without stepping into any function call step Execute one source line, stepping into function if the source line contains call to function finish Continue execution until the selected function returns Use current function if none is selected break Stop rogram execution at the line in the function selected Place the ca ret at the sta rt of source line on the function name Click the break button stop sign app ea rs next to the source line tbreak Set reakp oint enabled fo only one stop Same as the break button except that the reakp oint is automatically disabled after the �rst time it is hit delete Remove the reakp oint on the source line selected, the reakp oint numb er selected show brkpts Sho the current reakp oints (b oth active and inactive) stack Sho stack trace of functions called up Move up one level on the call stack down Move do wn one level on the call stack print Print the value of selected exp ression print Print the value of the object the selected exp ression is ointing to display Displa the value of selected exp ression in the displa windo w, up dating it every time execution stops undisplay Stop displa ying the value of the va riable in the displa windo If the selected exp ression is constant, it refers to the displa numb er asso ciated with an exp ression in the displa windo args Print the rguments of the selected frame show display Sho the names of currently displa ed exp ressions Debugging Interactive debugging Based on source co de, statement-level debugger Allo ws to discover values of va riables using their names in the source rogram, tracing their execution one statement at time The �les re compiled with the -g ag in e�ect Allo ws the inclusion of extra symb ol table info rmation in the bina ry �les Names and lo cations of all va riables Names of all functions and their rguments Data yp es of all objects decla red in the rogram ath names of the source co de �les used to compile the rogram xxgdb debugger Provides windo ws-o riented graphical user interface to gdb under the windo system Provides mouse selection fo va rious text commands Allo ws user to control rogram execution through reakp oints Consists of the follo wing windo ws File windo Displa ys the full pathname of the �le displa ed in the source windo Also displa ys the line numb er of the ca ret Source windo Contents of source �le Message windo Execution status and erro messages of xxgdb Command windo List of common gdb commands Commands invok ed clicking the left mouse button in the Dialog windo yping interface to gdb Displa windo Windo to displa va riable values opup windo ws Windo ws fo displa ying va riables ext selection exp ression selected clicking on the left mouse button Based on the resource delimiters to determine the set of cha racters that delimit exp ression Also ossible to select text holding do wn the left mouse button and dragging Pressing shift ey with left mouse button click displa ys the value of the va riable Scrollba Press left mouse button to scroll text fo rw rd Press right mouse button to scroll text backw rd Drag the middle mouse button to change the thumb osition of the text